Author's response to reviews

Title: Association between oral candidiasis and low CD4+ count among HIV positive patients in Hoima Regional Referral Hospital.

Authors:

   Martina Nanteza (nantezamartina@yahoo.com)
   Jayne B Tumusiime (jtusiime@musph.ac.ug)
   Joan Kalyango (nakayaga2001@yahoo.com)
   Arabat Kasangaki (arabat55@yahoo.com)

Version: 4  Date: 30 October 2014

Author's response to reviews: see over
Author’s covering letter for initial submission

Title: Association between oral candidiasis and low CD4+ count among HIV positive patients in Hoima Regional Referral Hospital.

Authors: Nanteza Martina, Jayne B Tusiime, Joan Kalyango, and Arabat Kasangaki

Version: Date: 30 October 2014

Comments: This article adds to the debate and knowledge regarding where a functional laboratory is not, what can a clinician use in determining HIV treatment and prognosis. It points out some salient prognostic features during patient examination which could facilitate staging of the disease and level of immunosuppression in the HIV/AIDS patient. It is important to know when and how to treat so as to have timely better outcomes.